Author's response to reviews

Title: Exploring obesogenic environments: Design and development of migrant obesogenic perception of the environment questionnaire (MOPE-Q)

Authors:

Maryam Delavari (mdela@deakin.edu.au)
Anders L. Sønderlund (a.l.sonderlund@exeter.ac.uk)
David Mellor (david.mellor@deakin.edu.au)
Mohammadreza Mohebbi (m.mohebbi@deakin.edu.au)
Boyd Swinburne (boyd.swinburn@auckland.ac.nz)

Version: 2
Date: 14 November 2013

Author's response to reviews:

The changes made to the manuscript include the addition of email addresses for all authors on the title page. We are aware that the editor also advised us to include an acknowledgements section, but all of the people who contributed to this article are listed as authors. There was also no source of funding for this article to list.